NOVA Experiment Status

All Experimenter’'s Meeting

Jaroslav Zalesak, Institute of Physics, Prague/FNAL

Dec 15, 2014




4 : '
Operations | - summary >

J BEAM:
e Beam for NOvA except Mon shutdown very good.

J NearDet:

e Another maintenance day onThu to keep less noisy.

* Except DSO scan very smooth running, no issues.

JFarDet:
° Physics with 14-diblocks running since Nov 20.

* We were bother by 1am crashes of Event Display till Fri.
* Minor DAQ issues when starting runs from scratch.

e Otherwise the detector was running stable.

J Control Room:

v" Satisfied with new Remote control room in Wichita.
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Data-Taking
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Hit rate recorded by FEB in Hz
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U Performed some DAQ tests to avoid
some issues when we start DAQ run

Partition 1 (status since Nov 20)

10/24/2014

maintenance, seeking good time slot. °

J We are preparing for a

sequences.

- NOvA Collaboration Meeting

Current status — Retrofitting: DONE!

= diblocks 1 — 14
" ALL COOLED, HYV at full gain.

" Running ~99% active,

Jaroslav Zalesal
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» Last week running with 94.4% uptime.

» We suffer with “lam Dispatcher issue” crashes since Mon through Fri, first shift crew
attempts to cure this problem caused other problems when runs started from scratch.

» Then we learned how to deal (the 1am crash did not actually require restarting the
system) and Fri we found hopefully culprit at least to avoid that.

» On Mon we used beam down time for DAQ tests which were longer than expected (we
lost some beam). Others down times where shorter in length, but restarting not so smoo‘t

Klaroslav Zalesak - AEM 12/15/2014
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/FD: POTs delivered/recorded
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> We recorded 8.37¢18 (from 8.80) POTs past weak, 2" best week in both!
» On Mon we tried to use beam shutdown to do maintenance for Far detector

Klaroslav Zalesak - AEM

» Past week efficiency 95.1% € good, weeks before even better 98.0 and 98.4%.

> As we run as much as possible we have very few possibilities to maintain detector.
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